Author’s response to reviews

Title: Risk factors associated with osteonecrosis of femoral head after internal fixation of femoral neck fracture: A systematic review and meta-analysis

Authors:

Jing-Li Xu (565193584@qq.com)
Zheng-Rong Liang (1399728486@qq.com)
Bing-Lang Xiong (1281195323@qq.com)
Qi-Zhao Zou (469497339@qq.com)
Tian-Ye Lin (1185527583@qq.com)
Yang Peng (284231697@qq.com)
Da Chen (dacheneisci@hotmail.com)
Qing-Wen Zhang (zh__qwen@163.com)

Version: 2 Date: 30 Aug 2019

Author’s response to reviews:

Dear editor:

Thank you for your valuable Suggestions. We have requested professionals to modify the language of the manuscript as required. And the declarations have been added as required and attached to the end of the manuscript. If there are any more mistakes, please let us know. Thank you.

Best regards!

Corresponding author
Name: Qingwen Zhang
E-mail: zh__qwen@163.com